Author’s response to reviews

Title: Efficacy of manipulation for non-specific neck pain of recent onset: design of a randomised controlled trial

Authors:

Andrew M Leaver (A.Leaver@fhs.usyd.edu)
Kathryn M Refshauge (K.Refshauge@fhs.usyd.edu.au)
Christopher G Maher (C.Maher@fhs.usyd.edu.au)
Jane Latimer (J.Latimer@fhs.usyd.edu.au)
Rob D Herbert (R.Herbert@fhs.usyd.edu.au)
Gwendolen Jull (g.jull@shrs.uq.edu.au)
James H McAuley (J.Mcauley@fhs.usyd.edu.au)

Version: Date: 26 February 2007

Author’s response to reviews:

Dear Editor

We re-submit this manuscript "Efficacy of manipulation for non-specific neck pain of recent onset: design of a randomised controlled trial." The following changes have been made
1. The sections of the title have been separated with a colon rather than a full stop
2. The author’s affiliations have been modified to the requested format (department, institute, city, country)
3. The abstract has been structured into Background, Methods/Design and Discussion sections
4. The Manuscript sections have been structured as instructed. There are no abbreviations, no figure legends, no tables and no additional files. Acknowledgement of sources of funding has been added. There are no additional acknowledgements

Yours sincerely

Andrew Leaver.